Author’s response to reviews

Title: Atypical gunshot injury to the right side of the face with the bullet lodged in the carotid sheath: a case report.

Authors:

    PETER A ONGOM (petongom@yahoo.co.uk)
    STEPHEN C KIJJAMBU (sckijjambu@chs.mak.ac.ug)
    JOSEPHAT JOMBWE (jombwe2001@yahoo.com)

Version: 3 Date: 25 November 2013

Author’s response to reviews: see over
25th November, 2013

THE EDITOR-IN-CHIEF,
JOURNAL OF MEDICAL CASE REPORTS

Dear Sir/Madam,

Re: Submission of revised Manuscript of a Case Report

On behalf of the authors, I hereby submit a second revised version of a manuscript for due consideration for publication in your journal. It is titled ‘Atypical gunshot injury to the right side of the face with the bullet lodged in the carotid sheath: a case report’. The authors are Peter A. Ongom, Stephen C. Kijjambu and Josephat Jombwe.

We have made revisions to the last revised manuscript following recommendations from the editor, so as to have it conform to the journal’s style. These include:

- Expurgated the name of the hospital where the patient was confined, so that the anonymity of the patient is not jeopardized
- Written Accident and Emergency Department in full; we had previously referred to A and E Department

Yours truly,

DR. PETER A. ONGOM
petongom@yahoo.co.uk
Corresponding Author